Author's response to reviews

Title: Expression of centromere protein F (CENP-F) associated with higher FDG uptake on PET/CT, detected by cDNA microarray, predicts high-risk patients with primary breast cancer

Authors:

Shigeto Ueda (syueda2000@yahoo.co.jp)
Nobuo Kondoh (nkondoh@dent.asahi-u.ac.jp)
Hitoshi Tsuda (htsuda@gmail.com)
Souhei Yamamoto (htsuda@ndmc.ac.jp)
Junji Yamamoto (iyamamot@ndmc.ac.jp)
Katsumi Tamura (abe@toko-pet.or.jp)
Jiro Ishida (iscida@nifty.com)
Yoshiyuki Abe (abe@toko-pet.or.jp)
Mikio Yamamoto (htsuda@ncc.ac.jp)
Hidetaka Mochizuki (mochizuki@ndmc.ac.jp)
Hideki Asakawa (hiddie_vr4@hotmail.com)
Kazuhiko Fukatsu (fukatsu@ndmc.ac.jp)
Takayuki Kobayashi (htsuda@ndmc.ac.jp)

Version: 10 Date: 10 December 2008

Author's response to reviews: see over
Dear The BioMed Central Editorial Production Team

Title: Expression of centromere protein F (CENP-F) associated with higher FDG uptake on PET/CT, detected by cDNA microarray, predicts high-risk patients with primary breast cancer
Journal: BMC Cancer
MS: 7216131472125441

We have correctly amended the revised version of our manuscript as you pointed out.

>Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

The list of authors in the manuscript is correct. We have reordered the list of authors in the submission system.

>Keywords, Word-counts, Table & Figure counts Please remove these.

We have removed Keywords, Word-counts, Table & Figure counts in the manuscript.

>(Abstract) -> Abstract. Please remove the brackets form each section heading

We have removed the brackets form each section heading.

>Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. Please remember to also update the Abstract details on the submission page.

We have correctly re-written the Abstract structure as Background, Methods, Results, Conclusions.

>Section headings Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT, Figure legends not Figure Legends.

We have correctly re-written all headings and sub-headings in the manuscript.
The manuscript refers to a supplementary figure, but this figure appears to have been uploaded as a normal figure file. Please amend the legend and all references to this figure in the manuscript text.

Figure order - The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9.

We have re-written a supplementary figure to Figure 4.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

We have checked our manuscript.